Author's response to reviews

Title: Automated dose dispensing service for primary health care patients - A systematic review

Authors:

Juha Sinnemäki (juha.sinnemaki@helsinki.fi)
Sinikka Sihvo (sinikka.sihvo@thl.fi)
Jaana Isojärvi (jaana.isojarvi@thl.fi)
Marja Blom (marja.blom@helsinki.fi)
Marja Airaksinen (marja.airaksinen@helsinki.fi)
Antti Mäntylä (antti.mantyla@fimea.fi)

Version: 5 Date: 12 December 2012

Author's response to reviews: see over
Dear editor,

Thank you for the comment regarding our manuscript "Automated dose dispensing service for primary health care patients - A systematic review". Changes are highlighted in the manuscript.

Comment

"Please ask the authors to change all occurrences of "under the ADD scheme" to "using the ADD scheme". Thanks."

The change has been made.

We hope that the revised version of our manuscript will be acceptable for publishing in Systematic Reviews.

Sincerely yours,

Juha Sinnemäki, M.Sc. (Pharm.)
Corresponding author
Division of Social Pharmacy
Faculty of Pharmacy
University of Helsinki
Viikinkaari 9 (P.O.Box 56)
00014 University of Helsinki
Finland
Tel: +358 500567183
e-mail: juha.sinnemaki@helsinki.fi